Piloting a Telephone Based Health Coaching Program for Pregnant Women: A Mixed Methods Study

Chris Rissel, Santosh Khanal, Jane Raymond, Vanessa Clements, Kit Leung & Michael Nicholl

Volume 17 • Number 7

Maternal and Child Health Journal

ISSN 1092-7875

Matern Child Health J DOI 10.1007/s10995-019-02735-2



MATERNAL AND CHILD HEALTH JOURNAL

Springer ISSN 1092-7875 • 10995 17(7) 1167-1338 (2013)





Your article is published under the Creative Commons Attribution license which allows users to read, copy, distribute and make derivative works, as long as the author of the original work is cited. You may selfarchive this article on your own website, an institutional repository or funder's repository and make it publicly available immediately.





Piloting a Telephone Based Health Coaching Program for Pregnant Women: A Mixed Methods Study

Chris Rissel¹ · Santosh Khanal¹ · Jane Raymond² · Vanessa Clements² · Kit Leung² · Michael Nicholl³

© The Author(s) 2019

Abstract

Objectives Get Healthy in Pregnancy (GHiP) is a telephone based lifestyle coaching service for pregnant women, in New South Wales, Australia. GHiP had two service options; a telephone-based health coaching program consisting of up to 10 calls and information only (including one call). This study sought to compare the outcomes of the two GHiP options, to determine the characteristics of women likely to use the service and to explore the feedback from women and health professionals. *Methods* A pragmatic stratified clustered randomised controlled trial was conducted. Two metro and three rural hospitals were randomised into health coaching or information only arms. Self-reported measures of height and weight and health behaviours (dietary and physical activity) were collected at baseline and 36 weeks gestation. Process evaluation included descriptive analysis of routine program data, and semi-structured interviews with participants and health professionals. *Results* Of 3736 women screened, 1589 (42.5%) were eligible to participate, and of those eligible, 923 (58.1%) were recruited. More women in the health coaching arm gained weight within the target range for their BMI at 36 weeks gestation (42.9%) compared with information only (31.9%). Women found GHiP to be useful and supportive and midwives and doctors said that it facilitated conversations about weight with pregnant women. *Conclusions for Practice* Telephone-based lifestyle programs integrated with routine clinical care show promise in helping pregnant women achieve healthy gestational weight gain, but in this case was not significantly different from one information telephone call. Strong positive feedback suggests that scaled-up service delivery would be well received.

Trial Registration ACTRN12615000397516 (retrospectively registered).

Keywords Health promotion · Health coaching · Gestational weight gain · Obesity · Maternal obesity

Chris Rissel Chris.Rissel@health.nsw.gov.au

Santosh Khanal khanal_santosh@hotmail.com

Jane Raymond Jane.raymond@health.nsw.gov.au

Vanessa Clements vanessa.clements@health.nsw.gov.au

Kit Leung Kit.leung@health.nsw.gov.au

Michael Nicholl michael.nicholl@health.nsw.gov.au

- ¹ NSW Office of Preventive Health, Liverpool, NSW, Australia
- ² NSW Ministry of Health, North Sydney, NSW, Australia
- ³ Department of Obstetrics and Gynaecology, Royal North Shore Hospital, St Leonards, NSW, Australia

Significance Statement

What is already known on this subject? Gestational weight gain for most women exceeds recommendations, and can lead to adverse health outcomes for women and their children. Interventions to support healthy weight gain are needed. What this study adds? Telephone coaching to support pregnant women to achieve healthy weight gain is well received by the women and supports health professionals in a professional dialogue about healthy weight. A series of telephone coaching calls led to a non-significantly higher proportion of women achieving their target weight gain compared to information only and one telephone call.

Background

A large body of evidence links excessive gestational weight gain (EGWG) with poor maternal and infant health outcomes (Gaillard et al. 2013; Mamun et al. 2010; Stotland et al. 2004), leading to an increased likelihood of postpartum obesity in both women (Mamun et al. 2010) and their children (Ludwig & Currie). In turn, these early adverse health outcomes lead to an increased risk of chronic disease later in life (Muktabhant et al. 2012). The Institute of Medicine (IOM) (Institute of Medicine (US) and National Research Council (US), 2009) has recommended a range for healthy weight gain for specific pre-pregnancy BMI categories based on the least risk of adverse perinatal outcomes (Stotland et al. 2006).

Gestational weight gain for most women exceeds the IOM recommendations (Institute of Medicine (US) and National Research Council (US), 2009), and in Australia, the prevalence of EGWG has been reported to be between 38% and 67% (Chung et al. 2013; de Jersey et al. 2012). The first national guidelines for antenatal care in Australia published in 2012 recommended that clinicians give women advice about appropriate weight gain during pregnancy in relation to their pre-pregnancy BMI (Australian Health Ministers' Advisory Council 2012). Women should receive interventions related to weight gain as early as possible in their pregnancy to help them make healthy choices that will maximise health outcomes for them and their babies.

A number of interventions (Thangaratinam et al. 2012) targeting weight gain in pregnancy have been developed but the literature reports that passive interventions such as leaflets and brochures which have limited effectiveness, or they have a face-to-face component which make them costly and resource-heavy when implemented on a large scale. A meta-analysis with 44 studies found that overall, the intervention group had a 1.42 kg reduction in gestational weight gain with any intervention compared with the control group, and with dietary intervention resulting in the largest reduction in maternal gestational weight gain (3.84 kg), and with improved pregnancy outcomes compared with other interventions (Thangaratinam et al. 2012). The quality of the studies varied considerably, with the quality of those studies addressing gestational weight gain considered moderate (Thangaratinam et al. 2012). To contribute to outcomes at a population level, the interventions need to be implemented widely and with sufficient intensity to improve individual risk behaviours (Cahalin et al. 2015). These outcomes are not achievable with most of the currently available interventions.

One potential intervention for promoting healthy weight gain during pregnancy is telephone-based health coaching.

Telephone-based health coaching delivered by gualified health professionals unrelated to the client's clinical care is a well-established care model for patients with chronic disease (Goode et al. 2012). Health coaching is designed to support clients through regular contact to achieve a healthier lifestyle through the use of behaviour change theory, and thereby also supporting the clinicians to support the overall health of their patients. An example is the Get Healthy Information and Coaching Service (GHS), which is a free telephone-based health coaching service that has been available to all people over 18 years of age in New South Wales (NSW), Australia, since 2009. People using this service can opt to receive information only or enrol in a 6-month coaching program during which they receive 10 individually tailored calls from university qualified coaches. The coaching calls aim to support participants in making sustained improvements in healthy eating, physical activity and achieving or maintaining a healthy weight. Participants who complete the 6-month coaching program have been shown to lose an average of 3.9 kg and reduce their waist circumference by 5.0 cm (O'Hara et al. 2012).

Using the GHS model, we developed the Get Healthy in Pregnancy (GHiP) service (Clements et al. 2016), with two service delivery options (6-months of health coaching and one call for information only). In this paper we compare the outcomes of the information only and telephone based coaching options of the GHiP service, determine the characteristics of pregnant women most likely to utilise the program and explored the feedback from women and health professionals related to usefulness of GHiP and program uptake and retention.

Materials and Methods

Study Design

This study employed a pragmatic stratified cluster randomised design, with stratification of participants by healthy weight and overweight/obese pre-pregnancy BMI and of hospitals by metropolitan and rural regions. Hospitals were randomised into the two GHiP options; telephone-based health coaching and information only. Given the evidence supporting appropriate weight gain during pregnancy for the health of the woman and her baby, a pure control group was not considered ethically acceptable.

Process evaluation was conducted which included an audit of the women who agreed to participate in the study but did not enrol into the program, and semi-structured interviews with women, midwives and the medical staff at the participating hospitals. The service provider that delivered the program collected routine program data including and the status of each participant i.e. continuing or withdrawn, the number of calls received by participants and the duration of each call.

Study Setting

The study was conducted in the antenatal clinics of five NSW public hospitals that account for approximately 10% of the total births in NSW. Three of the hospitals were rural (Orange Base, Lismore Base and Dubbo Base) and two were metropolitan (Liverpool and Blacktown). Two of the rural hospitals (Orange and Dubbo) were combined into one unit for randomisation because they are in the same local health district and their aggregated total number of births in 2012 was similar to the total of the other rural hospital in the study.

Ethics and Consent

The study was approved by the Human Research Ethics Committee of South Western Sydney Local Health District (HREC/14/LPOOL/131) and site specific approvals were obtained for each study site. Informed consent was obtained from the women by their attending midwives prior to the inclusion of the women in the study.

Inclusion and Exclusion Criteria

The inclusion criteria for the study were English speaking, 18 years and over, singleton pregnancy and gestation of 18 weeks or under. Women with medical conditions that may have impacted on their ability to participate in the study were identified to ensure safety to participate in the study. These women were either excluded from the study or were required to receive medical clearance prior to their inclusion (Clements et al. 2016). Women who were underweight at pre-pregnancy were excluded from the study.

Power Analysis/Sample Size

A total of 167 women were required in each of the health coaching and the information only arms arm to detect a 15% difference in the proportion of participants who gained weight within the target weight for their pre pregnancy BMI at a significance level of 0.05 with a power of 80%. We aimed to recruit 280 women in the health coaching arm allowing for an attrition rate of 60% and 235 women in the information only arm allowing for an attrition rate of 30%.

Recruitment Procedures

During the recruitment period (Sept 2014–Sept 2015) midwives screened every woman presenting for their first antenatal appointment to determine eligibility to be enrolled in the study. Women were screened for their eligibility and invited to participate in the study by the midwives during the women's first antenatal appointment at the hospital. The women self-reported their pre-pregnancy height and weight to the midwives.

Intervention Group Procedures

In the *information only arm*, midwives discussed gestational weight gain recommendations with the women and referred them to the GHiP service. When the women enrolled into the GHiP service, they were provided with one-off 20–30 min information only telephone call during which the women were provided information about appropriate weight gain in pregnancy and baseline data was collected. Evidence based written resources regarding weight gain in pregnancy were sent to the participants by post. The resources included factsheets about healthy eating, physical activity and weight gain during pregnancy and an information booklet about activities that could be undertaken to be healthy.

In the *telephone-based health coaching arm*, women received identical evidence-based written resources plus a journey booklet to record their progress and up to 10 health coaching calls (8 during pregnancy, 2 post-pregnancy). The journey booklet was designed as a resource for the women to track their progress through the program and was not used as one of the study evaluation tools. Baseline data was collected during the first call. The topics discussed during the calls are listed in Table 1.

Both GHiP arms commenced between 12 and 22 weeks gestation and all the phone calls were made by university qualified health professionals such as dieticians and exercise physiologists. Post study data was collected at 36 weeks gestation for both arms.

Data Collection

Data for the study was collected at screening (by midwives in antenatal clinics), pre-program (by health coaches during enrolment into GHiP for health coaching arm and during one-off information only call for information only participants) and at 36 weeks gestation (by health coaches or research assistant). The screening data, collected by the midwives at the hospital, included information regarding the inclusion and exclusion criteria for the study and the women's height and current weight. The women self-reported their pre-pregnancy weight in line with current routine maternity data collection across NSW. The pre-program and 36 week data collection included self-reported weight and validated self-reported dietary (Barr et al. 2008; McLennan and Podger 1998; Wright and Scott 2000), and physical activity measures (Australian Institute of Health and Welfare 2003; Smith et al. 2005). The items collected in the

Call No.	Stage of pregnancy	Content
1	18-21 weeks of gestation	Baseline data collection
1–6	1st and 2nd trimester	 Common problems of pregnancy, why they occur and what might help Common discomforts and what to do to relieve them (e.g. for nausea, have a snack and avoid greasy/spicy foods) Healthy eating: Australian Guide to Healthy Eating (5 food groups), snack ideas, myths (e.g. "eating for two") and recommended weight gain during pregnancy, important nutrients during pregnancy and lactation (e.g. folate, iodine, iron), foods to avoid Physical activity: benefits of exercise, tips and precautions, stretching exercises, type, intensity, frequency and duration, importance of variety, incidental daily activity
7–8	3rd trimester	 Feeding your baby and benefits of breastfeeding Healthy eating: portion sizes and serves, healthy plate, keeping motivated, food labels Physical activity: posture, back care and symptom cycle, pelvic floor exercises, physical activity towards the end of pregnancy
9–10	Post natal	 Healthy eating: eating out, tips for busy lives and easy meals Physical activity: relaxation Behaviour maintenance

Table 1 Gestation specific topic guide for the health coaches to tailor their conversations with the participants

dietary questionnaire include fruit and vegetable serves per day, cups of soft drinks per day and number of take away meals per week. Similarly the physical activity questionnaire included questions about walking and moderate and vigorous physical activity.

Semi-structured interviews with the women were conducted with a purposive sample of study participants at various stages of the program. Women were selected on the basis of their BMI status, the number of calls they received (0 calls, 1–3 calls completed, 4–7 calls completed and the 36 week call completed) and whether they were from metro or rural regions. The health professional semi-structured interviews were conducted with a convenient sample of midwives (N = 19) and medical practitioners (N = 5).

Data Analysis

A logistic regression of the combined screening and baseline data was conducted to determine the women who would be most likely to agree to participate in the study and enroll into the program. For the women who completed the 36 weeks gestation data collection, the differences between the two arms in mean weight change and changes in dietary scores at 36 weeks gestation and the proportions of pregnant women who achieved weight gain within the recommended range for their pre pregnancy BMI were conducted descriptively. All statistical analyses were done using IBM SPSS statistics v21 (IBM Inc, NY, USA). We followed the COREQ criteria for reporting qualitative research.

In this paper, we explored the women's participation and withdrawal from the study and the usefulness of GHiP to the midwives and the medical practitioners in the antenatal clinics. Other aspects of the process evaluation will be discussed in a subsequent paper. In the first few months of recruitment, an audit of randomly selected non-enrollees who agreed to participate in the study but did not subsequently enroll into GHiP was conducted. Semi-structured interviews were conducted with midwives, medical practitioners and with women who completed and those women who withdrew from GHiP at various stages. A thematic analysis of the interview data was conducted to identify the reasons for discontinuing from the program prior to the 36 week data collection point. The coding of the transcripts was undertaken and checked by independent coders, and themes and subthemes were extracted from the coded data. We included all of the elements of the RECORD checklist to match our study design.

Results

Overall 3736 women were screened across all the five hospitals and 1589 (42.5%) were eligible to participate in the study. Of the eligible women, 923 (58.1%) agreed to participate and were recruited into the study. A total of 322 women (34.9% of women recruited) enrolled into the program, of whom 89 (27.6%) completed the 36 week call. No differences in the withdrawal patterns between the health coaching and the information only arms were found (Fig. 1).

In the health coaching arm, 64.3% (27/42) of the women who reached the 36 week data collection point received all of the eight pre pregnancy calls. The number of calls received by the remaining women in the health coaching arm who reached the 36 week data collection point were evenly spread out (2 calls, n=2; 3 calls, n=2; 4 calls, n=4; 5 calls, n=2, 6 calls, n=3; 7 calls, n=2). In the health coaching arm, the first two calls were approximately 15 min in duration and the remaining calls were on average approximately



Fig. 1 The study flowchart

10 min each, and the information only call in the information only arm was on average 22 min in duration.

Program Uptake and Retention

No differences in the average age and pre pregnancy weight of women who were ineligible or eligible for the study were seen. Women attending rural hospitals were more likely to be ineligible (n=979, 74.1%) than women attending metro hospitals (n=1104, 45.9%) (p < 0.01). Attending the antenatal clinic after 18 weeks of gestation was the most common cause of ineligibility in both rural (n=937, 95.7%) and metro hospitals (n=835, 75.6%), although more so in rural hospitals (p < 0.01). Table 2 shows the proportion of women ineligible for the study by reason and study arm.

Of the eligible women, a higher proportion of women from the metro region (n = 602, 46.2%) than rural (n = 120, 35.1%) (p < 0.01) declined to participate in the study, but there were no differences in the recruitment rates when

compared by the study arms (health coaching, n = 291, 33.6%; information only, n = 407, 35.2%). The main reasons provided by the women for declining to participate in the study were non-interest in managing weight (n = 191, 27.4%) and lack of time (n = 148, 21.0%).

The logistic regression of the combined screening and baseline data showed that younger women from rural areas (p < 0.01) and women who had more previous pregnancies more than 20 weeks (p=0.04) were more likely to enrol into the program than other women. re-Pre-pregnancy BMI or the location of hospital on the recruitment into the study had no significant effect on outcomes. However, a higher proportion of women who were overweight or obese pre pregnancy compared with women of normal BMI were more likely to enrol into the program but withdraw before 36 weeks (p < 0.01). The demographics of women screened, recruited into the study and enrolled in the GHiP service were similar in both health coaching and information only arms (Table 3).

Program outcomes

The results for weight and health behaviour outcomes of the cohort of 89 women who completed 36 weeks of the program are provided in Table 4. The proportion of women who gained weight within the target range for their BMI between the health coaching (42.9%, 18/42) and the information only (31.9%, 15/47) arms, an 11.0% difference. However, this result was not statistically significant (p=0.29). Although positive changes were seen in both arms, the reductions in daily soft drink intake and weekly takeaway consumption were slightly higher for the health coaching arm than information only, as expected.

Process evaluation

In the audit of 26 randomly selected women who agreed to participate but did not enrol, 12 (46.2%) were successfully contacted. The phone numbers were incorrect or had been disconnected for five women who could not be contacted, and 12 women did not answer their phones.

Table 2 Reasons forineligibility to participate in thestudy		Health coaching (N=960, %)	Information only (N=1155, %)	Total (N=2115, %)
	Gestational age over 18 weeks	640 (66.7)	768 (66.5)	1408 (66.6)
	Non-English speaking	85 (8.9)	103 (8.9)	188 (8.9)
	Multiple pregnancy	22 (22.9)	21 (18.2)	43 (20.3)
	Pre-existing medical condition	36 (37.5)	25 (21.6)	61 (28.8)
	Underweight	93 (9.7)	80 (6.9)	173 (8.2)
	Under 18 years of age	38 (4.0)	29 (2.5)	67 (31.7)
	Unknown BMI	131 (13.6)	129 (11.2)	260 (12.3)

Table 3 Comparison of the demographics and weight status of women who were recruited, enrolled and withdrew by intervention arm

	Health coaching				Information only			
	Screened	Recruited	Enrolled in GHS—pre- program data	Withdrew from GHS after enrolling	Screening	Recruited	Enrolled in GHS—pre- program data	Withdrew from GHS after enrolling
Age (years \pm SD)	28.7 (5.7)	29.7 (6.5)	29.5 (4.7)	29.4 (4.3)	28.7 (6.3)	29.0 (5.1)	28.5 (5.2)	27.8 (5.3)
Average pre pregnancy weight (kg±SD)	61.1 (19.3)	65.0 (19.6)	62.3 (7.0)	71.4 (25.2)	58.8 (18.4)	61.1 (18.5)	58.6 (5.8)	62.1 (17.8)
% OWO (pre pregnancy)	47.4	50.0	60.2	64.0	40.8	50.0	50.3	55.2
% From rural hospitals	37.4	31.5	41.7	38.2	33.6	17.4	20.8	23.2

 Table 4
 Comparison between
 intervention and control arms of the changes in health behaviours and outcome measures

	Health coaching (N=42)	Information only (N=47)
Weight gain $(kg \pm SD)$	11.3 (±5.5)	13.6 (±7.7)
N (%) participants with weight gain within target range	18 (42.9)	15 (31.9)
Change in health behaviours (post-pre)		
No. times walking at least 30 min/week	$0.8 (\pm 2.6)$	1.3 (±3.1)
No. times of moderate to vigorous PA/week	$0.2 (\pm 1.6)$	$0.4 (\pm 1.6)$
Veg serves/day	0.5 (±1.3)	$0.2 (\pm 1.7)$
Fruit serves/day	$0.0 (\pm 1.0)$	$0.0(\pm 1.2)$
Cups soft drinks/day	$-0.4 (\pm 0.9)$	$-0.1(\pm 0.9)$
Times take away/week	$-0.4(\pm 1.0)$	$0.1 (\pm 1.0)$

One of the key themes that emerged in the context of study recruitment was that the women agreed to participate in the study primarily because they were asked by their midwife. In some instances, women also pointed out their interest in maintaining weight during and after pregnancy as their reasons for participation.

'I just thought I'll do it, because the midwife spoke to me about it and I just agreed to it.' (B5_P) 'Cause I was over the weight I wanted to be when I fell pregnant anyway. So I just wanted to make sure that I didn't put on ridiculous amounts of weight, that I'd have to then try and lose, once I had bub.' (C2_P)

The reasons consistently provided by the women for declining to participate in the study (audit of non-enrollees) or withdrawing after enrolling (semi-structured interviews) were lack of time and non-interest in weight management, healthy eating or exercise during pregnancy. Phone related issues such as disconnected phone numbers and missed or no calls were also provided by the women as reasons for non-enrolment or non-continuation.

"I found it difficult to keep up with the phone calls. I also have a three-years-old daughter as well, and just managing my time, and I was still working and all that, sort of, thing, so it was a bit difficult. So I just kept missing the phone calls." (D8_P)

The midwives and medical practitioners were generally positive about the program and suggested that GHiP facilitated their conversations about weight with women and helped them frame the topic positively as a routine conversation they had with all women, not just with women who were overweight or obese.

"Actually it's been good, because I can offer them something for it, so it's not just look you're overweight, you need to do something about it, it's this is something that can help you through it." (Y12_R) "It's been helpful because it [the service] gives another way of communicating with the patients and approaching them with their weight gain—it gives more support for what we are talking about in the clinic."(X6_DR)

Discussion

Our study was designed to examine the impact of GHiP service, however, the results have highlighted the issues with reach and uptake of a telephone based health coaching service for pregnant women. The findings suggest that the service was of some value for the women who completed it in helping them gain weight within the target range for their BMI but the inadequate sample size and low retention rates make it difficult to assure the validity of the outcomes. Also, GHiP facilitated a conversation between the health practitioners and the pregnant women about weight, a topic often considered too difficult and sensitive by antenatal care providers (Stotland et al. 2010).

In our study, a substantial proportion of women, more so in the rural hospitals than metro, did not attend the hospital until later than 18 weeks of their pregnancy making them ineligible to participate in the study. This attendance pattern was largely due to the maternity care models (Brock et al. 2014) that encourage community based care during early pregnancy. GHiP could be extended to pregnant women over 18 weeks of gestation to address this issue, however, the program would be less effective as women start gaining weight from second trimester onwards (Carmichael et al. 1997; Overcash et al. 2015). A recent report indicates that approximately half of pregnant women in NSW do not attend antenatal care until at least 14 weeks of gestation (Australian Institute of Health and Welfare 2015). Expanding the GHiP service through encouraging early referrals to GHiP from community based service providers such as general practitioners, practice nurses and early pregnancy services is essential.

A large proportion of eligible women agreed to participate in our study when asked by their midwives but did not follow through by enrolling into GHiP. Some women were possibly reluctant to decline a service offered by their midwife, even when they did not have any interest in participating in GHiP, as has been shown to be the case in other weight related interventions (Atkinson et al. 2013; Knight and Wyatt 2010). Low program uptake and retention rates, similar to our findings, have been reported for different types of weight related interventions for pregnant women in Australia (Chwah et al. 2016; Davis et al. 2012) and internationally (Knight and Wyatt 2010; Poston et al. 2013) that had face to face components. This pattern of participation implies that the telephone-based nature of GHiP is unlikely to have been the main reason for low participation, however, the finding also indicates that not all women prefer a telephone based service or via printed resources only and that other types of services will still be required.

Many pregnant women do not seem to prioritise weight as an issue during pregnancy which is most likely the underlying reason for the low uptake and high withdrawal from the program. Some of the women who suggested lack of time as their reason for withdrawing from the program may have done so due to non-interest as the calls would have taken only 10–15 min of their time. Similar findings of women not prioritising weight during pregnancy have been reported in previous studies (Gaudet et al. 2011; Shub et al. 2013). Another important finding from our study is that the women who were overweight or obese at pre-pregnancy were more likely to withdraw from the program than women with normal pre-pregnancy BMI. This withdrawal is a concern, as overweight and obese women have an increased risk of adverse pregnancy outcomes (Athukorala et al. 2010). These findings suggest that approaches to increase the knowledge and awareness about weight gain during pregnancy are required alongside weight management programs.

Our study was not fully able to assess the effectiveness of GHiP, as originally intended, because of the low level of program uptake and retention and the disproportionally higher withdrawal rates for overweight and obese women. Other limitations were a pure control group was lacking and the pre-pregnancy and post-program weight data was self reported by the women. Recall errors in self-estimated pre-pregnancy weight (McClure et al. 2011) may have led to incorrect classification of women into the normal BMI and overweight or obese categories which would have resulted in those women being given the wrong target range for weight gain. However, this approach to data collection is in line with pragmatic nature of our trial as weight gain advice is given by midwives in NSW based on self-reported pregnancy BMI. Future research should aim to compare higher and lower intensity variations of the service to assess relative impacts. Stronger qualitative data may also help explain how to maximise the intervention effect.

Despite the above limitations, our study suggests that GHiP is a valuable cost-effective addition to routine antenatal care that can be easily incorporated into the current care model for supporting women in NSW to achieve a healthy gestational weight gain, raise awareness of the importance of healthy weight gain in pregnancy and help facilitate conversations between health professionals and pregnant women in line with antenatal care guidelines. Feedback from the women about GHiP was generally positive and indicated a level of acceptance among users.

Telephone based health coaching does not suit all pregnant women which makes the conversations between the health professionals and pregnant women even more important. One of the most important outcomes of integrating the GHiP program with the hospital antenatal services has been the facilitation of the conversations about weight between the pregnant women and their midwives and the medical practitioners. To help pregnant women manage their weight in pregnancy, these type of conversations should be happening not only at the hospital antenatal clinics but also earlier during the pregnancy. Early pregnancy care providers such as GPs and community nurses should discuss weight gain during pregnancy with the pregnant women. Programs such as GHiP can offer an accessible and a cost-effective referral pathway for the women these health practitioners see a need for additional support to manage their weight during pregnancy.

Acknowledgements The authors would like to thank Michelle Maxwell for her contribution to the planning of the study and development of the GHiP program, the midwives and medical officers at the five study sites who recruited women to the service, the clinic coordinators for the local management of the trial, Healthways staff for providing the coaching service and the GHiP Expert Advisory Group for clinical advice. The authors also acknowledge Praveena Gunaratnam for working on the original proposal and ethics application, Kit Leung and Vicky Aouad for coordinating the intervention and Karin Taylor for her contribution to the reporting as part of their Public Health Officer training with the NSW Ministry of Health.

Author Contributions All authors contributed to the intellectual planning of the study, intellectual analysis of the data and writing of the paper. SK, JR, VC and KL contributed to conducting the project. SK and CR contributed to the analysis of the data.

Data Availability The datasets generated and/or analysed during the current study are not publicly available as they contain identifiers routinely collected by the midwives and the health coaches. The data that support the findings of this study are also available from Health ways Australia, the external service provider that delivers the GHiP program. However, restrictions apply to the availability of these data, and so are not publicly available. All requests for data will have to be made to the Get Healthy Service Manager at the NSW Office of Preventive Health.

Compliance with Ethical Standards

Conflict of interest None of the authors have any competing interests.

Open Access This article is distributed under the terms of the Creative Commons Attribution 4.0 International License (http://creativecommons.org/licenses/by/4.0/), which permits unrestricted use, distribution, and reproduction in any medium, provided you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons license, and indicate if changes were made.

References

- Athukorala, C., Rumbold, A. R., Willson, K. J., & Crowther, C. A. (2010). The risk of adverse pregnancy outcomes in women who are overweight or obese. *BMC Pregnancy and Childbirth*, 10(1), 1.
- Atkinson, L., Olander, E. K., & French, D. P. (2013). Why don't many obese pregnant and post-natal women engage with a weight management service? *Journal of Reproductive and Infant Psychology*, 31(3), 245–256.
- Australian Health Ministers' Advisory Council (2012) Clinical practice guidelines: Antenatal care Module 1.
- Australian Institute of Health and Welfare. (2003). *The active australia survey: A guide and manual for implementation, analysis and reporting:* Australian Institute of Health and Welfare.
- Australian Institute of Health and Welfare. (2015). Australia's mothers and babies 2013—in brief. Retrieved from Canberra.
- Barr, M., Baker, D., Gorringe, M., & Fritsche, L. (2008). NSW population health survey: Description of methods. Sydney: NSW Department of Health.
- Brock, E., Charlton, K. E., & Yeatman, H. (2014). Identification and evaluation of models of antenatal care in Australia—a review of the evidence. *Australian and New Zealand Journal of Obstetrics* and Gynaecology, 54(4), 300–311.

- Cahalin, L. P., Kaminsky, L., Lavie, C. J., Briggs, P., Cahalin, B. L., Myers, J., et al. (2015) Development and implementation of worksite health and wellness programs: A focus on non-communicable disease. *Progress in Cardiovascular Diseases.* 58(1), 94–101. https://doi.org/10.1016/j.pcad.2015.04.001.
- Carmichael, S., Abrams, B., & Selvin, S. (1997). The pattern of maternal weight gain in women with good pregnancy outcomes. *American Journal of Public Health*, 87(12), 1984–1988.
- Chung, J. G., Taylor, R. S., Thompson, J. M., Anderson, N. H., Dekker, G. A., Kenny, L. C., et al. (2013). Gestational weight gain and adverse pregnancy outcomes in a nulliparous cohort. *European Journal of Obstetrics & Gynecology and Reproductive Biology*, 167(2), 149–153. https://doi.org/10.1016/j.ejogrb.2012.11.020.
- Chwah, S. R., Reilly, A., Hall, B., O'Sullivan, A. J., & Henry, A. (2016). Engagement with and outcomes of a midwifery-led intervention group for pregnant women of high body mass index. *Obstetric Medicine: The Medicine of Pregnancy*, 9, 120–125.
- Clements, V., Leung, K., Khanal, S., Raymond, J., Maxwell, M., & Rissel, C. (2016). Pragmatic cluster randomised trial of a free telephone-based health coaching program to support women in managing weight gain during pregnancy: The Get Healthy in Pregnancy Trial. BMC Health Services Research, 16(1), 454.
- Davis, D. L., Raymond, J. E., Clements, V., Adams, C., Mollart, L. J., Teate, A. J., & Foureur, M. J. (2012). Addressing obesity in pregnancy: The design and feasibility of an innovative intervention in NSW, Australia. *Women and Birth*, 25(4), 174–180.
- de Jersey, S. J., Nicholson, J. M., Callaway, L. K., & Daniels, L. A. (2012). A prospective study of pregnancy weight gain in Australian women. *The Australian and New Zealand Journal of Obstetrics and Gynaecology*, 52(6), 545–551. https://doi.org/10.1111/ ajo.12013.
- Gaillard, R., Durmus, B., Hofman, A., Mackenbach, J. P., Steegers, E. A., & Jaddoe, V. W. (2013). Risk factors and outcomes of maternal obesity and excessive weight gain during pregnancy. *Obesity*, 21(5), 1046–1055. https://doi.org/10.1002/oby.20088.
- Gaudet, L. M., Gruslin, A., & Magee, L. A. (2011). Weight in pregnancy and its implications: What women report. *Journal of Obstetrics and Gynaecology Canada*, 33(3), 227–234.
- Goode, A. D., Reeves, M. M., & Eakin, E. G. (2012). Telephonedelivered interventions for physical activity and dietary behavior change: An updated systematic review. *American Journal of Preventive Medicine*, 42(1), 81–88. https://doi.org/10.1016/j.amepr e.2011.08.025.
- Institute of Medicine (US) and National Research Council (US). (2009). Committee to reexamine IOM pregnancy weight guidelines. In K. M. Rasmussen & A. L. Yaktine (Eds.), Weight gain during pregnancy: Reexamining the guidelines. Washington: Institute of Medicine (US) and National Research Council (US).
- Knight, B. A., & Wyatt, K. (2010). Barriers encountered when recruiting obese pregnant women to a dietary intervention. *Nursing Times*, 106(32), 20–22.
- Ludwig, D. S., & Currie, J. The association between pregnancy weight gain and birthweight: A within-family comparison. *The Lancet*, 376(9745), 984–990. https://doi.org/10.1016/S0140 -6736(10)60751-9.
- Mamun, A. A., Kinarivala, M., O'Callaghan, M. J., Williams, G. M., Najman, J. M., & Callaway, L. K. (2010). Associations of excess weight gain during pregnancy with long-term maternal overweight and obesity: Evidence from 21 years postpartum follow-up. *The American Journal of Clinical Nutrition*, 91(5), 1336–1341. https://doi.org/10.3945/ajcn.2009.28950.

- McClure, C. K., Bodnar, L. M., Ness, R., & Catov, J. M. (2011). Accuracy of maternal recall of gestational weight gain 4–12 years after delivery. *Obesity*, 19(5), 1047–1053.
- McLennan, W., & Podger, A. S. (1998). National nutrition survey: Nutrient intakes and physical measurements. Canberra: Australian Bureau of Statistics.
- Muktabhant, B., Lumbiganon, P., Ngamjarus, C., & Dowswell, T. (2012). Interventions for preventing excessive weight gain during pregnancy. *The Cochrane Database of Systematic Reviews*, 4, CD007145. https://doi.org/10.1002/14651858.CD007145.pub2.
- O'Hara, B. J., Phongsavan, P., Venugopal, K., Eakin, E. G., Eggins, D., Caterson, H., et al. (2012). Effectiveness of Australia's get healthy information and coaching service(R): Translational research with population wide impact. *Preventive Medicine*, 55(4), 292–298. https://doi.org/10.1016/j.ypmed.2012.07.022.
- Overcash, R. T., Hull, A. D., Moore, T. R., & LaCoursiere, D. Y. (2015). Early second trimester weight gain in obese women predicts excessive gestational weight gain in pregnancy. *Maternal* and Child Health Journal, 19(11), 2412–2418.
- Poston, L., Briley, A. L., Barr, S., Bell, R., Croker, H., Coxon, K., et al. (2013). Developing a complex intervention for diet and activity behaviour change in obese pregnant women (the UPBEAT trial); assessment of behavioural change and process evaluation in a pilot randomised controlled trial. *BMC Pregnancy and Childbirth*, 13(1), 1.
- Shub, A., Huning, E. Y., Campbell, K. J., & McCarthy, E. A. (2013). Pregnant women's knowledge of weight, weight gain, complications of obesity and weight management strategies in pregnancy. *BMC Research Notes*, 6(1), 1.
- Smith, B. J., Marshall, A. L., & Huang, N. (2005). Screening for physical activity in family practice: Evaluation of two brief assessment tools. *American Journal of Preventive Medicine*, 29(4), 256–264.
- Stotland, N., Cheng, Y., Hopkins, L., & Caughey, A. (2006). Gestational weight gain and adverse neonatal outcome among term infants. *Obstetrics & Gynecology*, 108(3), 635–643. https://doi. org/10.1097/01.AOG.0000228960.16678.bd.
- Stotland, N., Gilbert, P., Bogetz, A., Harper, C., Abrams, B., & Gerbert, B. (2010). Preventing excessive weight gain in pregnancy: How do prenatal care providers approach counseling? *Journal of Women's Health*, 19(4), 807–814.
- Stotland, N., Hopkins, L., & Caughey, A. (2004). Gestational weight gain, macrosomia, and risk of cesarean birth in nondiabetic nulliparas. *Obstetrics & Gynecology*, 104(4), 671–677. https://doi. org/10.1097/01.AOG.0000139515.97799.f6.
- Thangaratinam, S., Rogozińska, E., Jolly, K., Glinkowski, S., Roseboom, T., Tomlinson, J. W., et al. (2012). Effects of interventions in pregnancy on maternal weight and obstetric outcomes: Metaanalysis of randomised evidence. *BMJ*. https://doi.org/10.1136/ bmj.e2088.
- Wright, J., & Scott, J. (2000). The fat and fibre barometer, a short food behaviour questionnaire: Reliability, relative validity and utility. *Australian Journal of Nutrition and Dietetics*, 57(1), 33–39.

Publisher's Note Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.